Author's response to reviews

Title: Early eczema and the risk of childhood asthma: a prospective, population-based study

Authors:

Marit Saunes (marit.saunes@ntnu.no)
Torbjørn Øien (torbjorn.oien@ntnu.no)
Christian K Dotterud (christian.dotterud@gmail.com)
Pål R Romundstad (pal.romundstad@ntnu.no)
Ola Storror (ola.storror@ntnu.no)
Turid L Holmen (turid.lingaas.holmen@ntnu.no)
Roar Johnsen (roar.johnsen@ntnu.no)

Version: 4 Date: 12 September 2012

Author's response to reviews: see over
Re-submission of revised manuscript: “Early eczema and the risk of childhood asthma; a prospective, population-based study”

Dear Editor,

We appreciate the opportunity to re-submit a revised version of our manuscript. We have thoroughly gone through the manuscript. Minor changes have been made in the abstract. Some spellings and orthographics have been corrected. Hopefully, the manuscript have improved.

Yours sincerely,

Marit Saunes (sign)
author
Department of Public Health and General Practice, NTNU
N-7489 Trondheim
Norway
Phone: +47 73598876
Fax: +47 73597577
E-mail: marit.saunestnu.no

Torbjørn Øien (sign)
author
Ola Storrø (sign)
author
Christian K Dotterud (sign)
author

Pål R. Romundstad (sign)
author
Turid Lingaas Holmen (sign)
author

Roar Johnsen (sign)
author